Author’s response to reviews

Title: Relations between the Crowe Classification and the 3D Femoral Head Displacement in Patients with Developmental Dysplasia of the Hip

Authors:

Rongshan Cheng (chengrongshan@sjtu.edu.cn)

Henghui Zhang (medzhh@163.com)

Willem Kernkamp (wakernkamp@gmail.com)

Jingmao Zheng (635587323@qq.com)

Kerong Dai (krdai@163.com)

Yifei Yao (yaoyifeiyyf@gmail.com)

Liao Wang (wang821127@163.com)

Tsung-Yuan Tsai (tytsai@sjtu.edu.cn; evanny.tsai@gmail.com)

Version: 3 Date: 17 Sep 2019

Author’s response to reviews:

Author’s response to reviews

Title: Relations between the Crowe Classification and the 3D Femoral Head Displacement in Patients with Developmental Dysplasia of the Hip

Authors:

Rongshan Cheng (chengrongshan@sjtu.edu.cn)

Henghui Zhang (medzhh@163.com)

Willem Alexander Kernkamp (wakernkamp@gmail.com)

Jingmao Zheng (635587323@qq.com)

Kerong Dai (krdai@163.com)

Yifei Yao (yaoyifeiyyf@gmail.com)
1. Please update the first line of the Methods to remove the "blinded for review" statement.
Answer: We have removed it.

2. We ask that you include a list of abbreviations (before the declarations section) for reference and ease of reading. All abbreviations should still be defined in the text at first use.
Answer: We have added abbreviations as you have commented, thank you.

3. In Funding, please state whether or not the funding body played any roles in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.
Answer: Considering the Editor’s comment, we have added the explanation of the role of the funding body in the revised version.

4. In the Authors Contributions section each author should also be specifically mentioned by their unique initials rather than their full name. Also, please include a statement in the Authors' contributions section to the effect that all authors have read and approved the final manuscript, and ensure that this is the case.
Answer: We have revised it as you have commented, thank you.

5. Upon resubmission, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Please ensure the document is in the final form for publication; please upload only files that are to be published and remove any files that should not form part of the final manuscript (e.g. covering letters, responses to reviewers, etc). If you wish to respond to these revision requests, please insert the information in to the designated input box only.
Answer: We have uploaded a clean version, thank you.